Accelerator Status Report
Run Schedule Jan. 14, 2009 to March 16, 2009
· Injector energy 52.66 MeV

· Linac Setup 4.734 MeV

· Pass setup: 5/5/5
· From Thursday evening to Monday morning the accelerator was running alright except some minor problems (RF, iocs and orbits). SAD started at 0700 Monday, March 16. 
· The accelerator was down from early Monday (3/9) to Thursday morning (3/12) due to vacuum issues of cryo module 1L04. By Thursday noon the valves in 1L04 region were opened.
· In order to have a stable machine the accelerator was reconfigured from 5.9GeV to 4.7GeV with zones 1L03, 1L04 and 1L05 off.

· Thursday (3/12) swing shift there was a VESDA alarm in Hall B. The crew followed the emergency response procedure. There was no real fire.

· Friday (3/13) after a power supply for iocin1 was replaced the chopper slits positions were not restored. A home reset was needed to correct the problem.

· Saturday (3/14) swing shift the orbits in the main machine drifted twice. The locks corrected the problems with tune beam to the hall dumplettes. Do not know why the orbits drifted. 
· Two adjacent ODH monitors were bypassed in both NL and SL when in beam permit.

· The accelerator shutdown process went well Monday (3/16).
· The accelerator was affected by some box supplies and the vacuum at 1L04. Arc2, Arc4 and RSEP9A each tripped off a couple of times. EES replaced the faulty parts. The vacuum at 1L04 started to deteriorated Sunday and was getting worse Monday. Cryo module 1L04 is at about room temperature. The warm window of cavity 1L04-5 may have a leak. A pump was connected to it to help pump. The accelerator has been down since early Monday morning.
Hall A
E06-014    4.7GeV     5 Pass
Measurement of Single Target-Spin Asymmetry
· Hall A beam was fine.
· The hall was changed back to pass 4. 

Hall B
Eg1-DVCS    4.7 GeV    5pass

· Hall B beam position was off when the main machine orbits/energy drifted Saturday, which tripped off its detector.
· No problem with Hall B beam

Hall C

E07-003-SANE     4.7 GeV    5 Pass 
· Friday owl shift (3/13) Hall C beam would not transport beyond 3C18 after a QE measurement.
After MQE3M02 was turned off beam transported the hall dump. 
· Friday swing shift (3/13) the last three BPMs of Hall C line was not stable. The crew identified the cause of the problem: Hall C slow raster. The slow raster was set to a condition such that the beam was affected in Y direction. 
· The hall was changed back to pass 5. Hall C chicane was adjusted and beam is in “straight” mode. The chicane dipoles BE and BZ are off. The field of the target coil is parallel to beam so it does not bend beam (beam goes to the regular dump). 
